

## **Author's response to reviews**

**Title:** Focusing on fast food restaurants alone underestimates the relationship between neighborhood deprivation and exposure to fast food in a large rural area

### **Authors:**

Joseph R Sharkey ([jrsharkey@srph.tamhsc.edu](mailto:jrsharkey@srph.tamhsc.edu))  
Cassandra M Johnson ([cjohnson@srph.tamhsc.edu](mailto:cjohnson@srph.tamhsc.edu))  
Wesley R Dean ([wdean@srph.tamhsc.edu](mailto:wdean@srph.tamhsc.edu))  
Scott A Horel ([sahorel@srph.tamhsc.edu](mailto:sahorel@srph.tamhsc.edu))

**Version:** 3 **Date:** 23 January 2011

### **Author's response to reviews:**

We made all formatting changes as requested.